Author's response to reviews

Title: Efficacy, predictability, and safety of small incision lenticule extraction: prospective cohort study

Authors:

Jae Ryun Kim (oindiano99@gmail.com)
Hyung Bin Hwang (leoanzel@catholic.ac.kr)
Su Joung Mun (moons-6@daum.net)
Young Taek Chung (eyeegreen@eyeegreen.com)
Hyun Seung Kim (Sara514@catholic.ac.kr)

Version: 2 Date: 26 March 2014

Author's response to reviews: see over
Dear Editor-in-Chief,

Thanks for giving opportunity to revise the article.

The design of this study is not clinical trials but prospective cohort study. We changed the title from “prospective clinical study” to “prospective cohort study” to avoid misunderstanding. The “Acknowledgements” representing source of funding also was added in manuscript. We look forward to your review of our work soon.

Yours faithfully,

Hyun Seung Kim, MD

Department of Ophthalmology and Visual Science, College of Medicine

The Catholic University of Korea

#62 Yeouido-dong, Yeongdeungpo-gu, Seoul, Korea 150-713

Tel: 82-2-3779-1243

Fax: 82-2-761-6869

E-mail: sara514@catholic.ac.kr